Author's response to reviews

Title: Strategies to improve the efficiency and utility of Multidisciplinary Team meetings in cancer care: a survey study

Authors:

Rozh T Jalil (r.jalil@imperial.ac.uk)
Benjamin W Lamb (benjamin.lamb@imperial.ac.uk)
Nick Sevdalis (n.sevdalis@imperial.ac.uk)
Charles Vincent (c.vincent@imperial.ac.uk)
James SA Green (james.green@bartshealth.nhs.uk)

Version: 2
Date: 19 December 2013

Author's response to reviews: see over
Dear Professor Christopher Morrey,

Please find attached our revised manuscript, “Strategies to improve the efficiency and utility of Multidisciplinary Team meetings in cancer care: a survey study” (MS: 3905509431159240). We have made changes to the original manuscript as requested by Armee Valencia. Specific points of revision are detailed below.

Statement of ethics: a statement of ethics board review has been included in the methods section.

Formatting: The structure of the abstract has been revised in accordance with journal requirements. A conclusions section has been added after the discussion.

Thank you for the opportunity to revise the original manuscript. We would be grateful if you would consider the manuscript for publication in BMC Health Services Research.

I look forward to hearing from you.

Yours sincerely,

Mr BW Lamb, MBBS, MA, MRCS, PhD.